Author’s response to reviews

Title: Medullary Thyroid Carcinoma with double negative calcitonin and CEA: A case report and update of literature review

Authors:

Claudio Gambardella (claudiog86@hotmail.it)

Chiara Offi (chiara.o@live.com)

Guglielmo Clarizia (clarizia.guglielmo@libero.it)

Roberto Maria Romano (robertomaria.romano@gmail.com)

Immacolata Cozzolino (coimma73@gmail.com)

Marco Montella (montella.marco19@gmail.com)

Rosa Maria Di Crescenzo (crescenzo-rose.dicrescenzo@libero.it)

Massimo Mascolo (massimo.mascolo@unina.it)

Angelo Cangiano (angicangi@gmanil.com)

Sergio Di Martino (sergiodm@libero.it)

Giancarlo Candela (giancarlo.candela@unicampania.it)

Giovanni Docimo (giovanni.docimo@unicampania.it)

Version: 3 Date: 21 Sep 2019

Author’s response to reviews:

Editor Comments:

1. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Thanks the Editor for the suggestion. The Authors’names were modified accordingly.
2. Please confirm if RRM in the Authors' Contributions section refers to Roberto Maria Romano.

Yes RRM is Roberto Maria Romano

3. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

The suggested sentence was added to the text

4. It has come to our attention that within your manuscript there is significant text overlap with other publications, particularly:

Page 2 lines 18-24, page 3 lines 20-24, page 4 lines 1-5, page 4 lines 10-19, page 7 lines 9-17, page 8 lines 8-25, page 9 lines 4-11, page 9 lines 20-25, page 10 lines 1-7, page 10 lines 16-20, page 11 lines 3-10, table 1:


Page 7 lines 19-24:


While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please rephrase this section in your own words to avoid direct overlap (please note, we cannot accept direct textual overlap with any previously published works even if the authors of the work are yourselves). Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

According to the reviewer suggestion the sentences were reformulated. The Table 1 is similar to the one of Gambardella, Claudio, et al. "Calcitonin negative Medullary Thyroid Carcinoma: a challenging diagnosis or a medical dilemma?." BMC endocrine disorders 19.1 (2019): 45 because is simply an update of the Literature. Nevertheless I tried to modify something

5. Please remove the CARE and response to reviewer additional files, as they are no longer required.
The suggested material was not included in the submission

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The manuscript is a single final clean version as requested